Author's response to reviews

Title: Is health related quality of life associated with the risk of low-energy wrist fracture? A case control study.

Authors:

Gudrun Rohde (gudrun.rohde@sshf.no)
Anne M Mengshoel (a.m.mengshoel@medisin.uio.no)
Astrid K Wahl (a.k.wahl@medisin.uio.no)
Torbjorn Moum (torbjorn.moum@medisin.uio.no)
Glenn Haugeberg (glenn.haugeberg@sshf.no)

Version: 10 Date: 1 July 2009

Author's response to reviews:

Once again I have tried to do an edition in the abstract. I hope it is correct this time.